# KRANNERT GRADUATE SCHOOL OF MANAGEMENT

Purdue University
West Lafayette, Indiana

Uncertainty and the Likelihood of Entry: An Empirical Assessment of the Moderating Role of Irreversibility

by

Tim Folta
Douglas R. Johnson
Jonathon O'Brien

Paper No. 1143 Date: May 2001

Institute for Research in the Behavioral, Economic, and Management Sciences

# **Uncertainty and the Likelihood of Entry:**

# An Empirical Assessment of the Moderating Role of Irreversibility

Timothy B. Folta\*

Douglas R. Johnson

Jonathan O'Brien

#### All from

Krannert Graduate School of Management Purdue University West Lafayette, IN 47907-1310 Phone: 765 494 9252 Fax: 765 494 9658

Foltat@mgmt.purdue.edu
Johnsond@mgmt.purdue.edu

# May 2001

\* Authors are listed alphabetically. Correspondence should go to this author.

The authors thank seminar participants at Purdue University, the 2001 Strategic Management Conference hosted by the University of Utah and Brigham Young University, and the 2001 SRF Conference at Northwestern University.

		ς.

# **Uncertainty and the Likelihood of Entry:**

# An Empirical Assessment of the Moderating Role of Irreversibility

#### Abstract

Real options models have emerged as a compelling tool for understanding market entry decisions involving sunkness, or irreversibility. In contrast to neoclassical models of investment, real options models consider that uncertainty has a direct effect on the willingness to invest, and that the effect of uncertainty is intensified as the irreversibility of investment increases. Although the effect of uncertainty on investment has been documented, the theoretical interaction between uncertainty and irreversibility has not been empirically demonstrated because the existing literature has tended to assume that all investments are equally irreversible. Our paper addresses this issue and advances the literature on real options theory in the following three ways. First, we empirically validate the proposition that uncertainty is moderated by the irreversibility of the required investment. Second, we relate the irreversibility of the entry decision to a firm's corporate strategy, and demonstrate that the effect of uncertainty is moderated by how related that industry is to the firm's current portfolio of business segments. Finally, our results provide evidence regarding the tradeoff between the option to grow and the option to wait – the effect of uncertainty on entry varies in the presence of growth options. (193 words)

#### 1.0 Introduction

In recent years, real options models have emerged as a compelling tool for evaluating market entry decisions (Dixit, 1989). Unlike traditional discounted cash flow approaches, real options models recognize that firms face certain opportunity costs when investments are irreversible. In particular, they bear the opportunity cost of foregoing the *option to wait* for additional information that may affect their decision. The magnitude of this opportunity cost is a function of both the level of uncertainty the firm faces as well as the degree of irreversibility associated with the investment.<sup>1</sup> As both uncertainty and irreversibility increase, the value of the option to wait and the incentives to delay entry increase. Thus, uncertainty is expected to have a direct effect on entry if it involves some irreversibility. If entry decisions can be costlessly reversed, uncertainty should have no direct effect on entry.

This interaction between uncertainty and irreversibility lies at the heart of this emerging literature on real options that applies not only to entry decisions, but to the broader study of investment (Dixit and Pindyck, 1994). Over the past decade, a significant body of empirical research has developed assessing the direct effect of uncertainty on investment and entry from a real options perspective (Carruth, et al., 2000). However, these studies have, for the most part, only looked at this phenomenon at the industry level of analysis. This is unfortunate for three reasons. First, by regressing uncertainty on industry (or even firm) investment levels, these studies make the assumption that all investments are equally irreversible. In the context of industry entry, this leaves virtually untested the central hypothesis that the probability of entry is determined by the interaction between uncertainty and irreversibility. Second, as Dixit and

<sup>1</sup> In employing the term "uncertainty", we are referring to exogenours uncertainty, or the volatility of the stochastic process determining the returns from investment, consistent with Dixit and Pindyck (1994).

Pindyck (1994) argue, option models do not describe the level of investment *per se*, but identify the threshold at which firm-level investment should occur. As such, investigating simple relationships between rates of investment and uncertainty cannot directly test the models. Third, testing for the effect of uncertainty on total industry investment ignores the possibility that investment thresholds (entry trigger points) may differ across firms. Unless it is explicitly considered in empirical models, heterogeneity across firm-specific investment thresholds may bias or compromise efforts to test the predictions from real option theory.

This study seeks to partially address these issues by examining the combined effects of irreversibility and uncertainty on entry into new industries. We consider that investment thresholds may differ across industries because entry into some industries may require different levels of irreversible investment in physical assets, such as equipment or buildings, or intangible assets, such as reputation or technical ability. We also consider that investment thresholds may differ across firms. Borrowing from a rich literature examining entry from the perspective of the diversifying firm (Montgomery & Hariharan, 1991; Penrose, 1959), we argue that the level of irreversibility of entry into a particular industry will be contingent upon a firm's ability to redeploy assets into its current portfolio of business activities. Firms diversifying into industries more related to their current portfolio of businesses will be making more reversible investments, and the effect of uncertainty on those firms' entry decisions will be less.

Using the Compustat II (Business Segment) database, we identified 12,127 instances where firms entered new business segments during the period from 1980 to 1999. These entries were then classified into 48 different industry groups using a classification proposed by Fama and French (1997). Using Fama and French's stock market returns for these 48 industry groups, we measured the level of time-varying uncertainty facing each industry. These measures of

uncertainty were used in conjunction with other data to assess both the direct effect of uncertainty on entry as well, as the moderating effect of irreversibility on the relationship between uncertainty and entry.

The remainder of this paper is organized as follows: section 2 reviews the theoretical issues; section 3 discusses past empirical research; section 4 presents the model specification, the data sources, and the measures; section 5 presents the empirical findings; and section 6 discusses the implications of our study as well as directions for future research.

# 2.0 Theoretical Background

# 2.1 Neoclassical Investment Theory Versus Option Theory

Traditional investment theory predicts that a firm will expand into an industry when the net present value (NPV) of the expected cash flows from expansion is equal to or greater than zero. Underlying this prediction are two implicit assumptions that do not always match with reality (Pindyck, 1991). One underlying assumption is that investments are fully *reversible* - they can be recovered and costlessly redeployed if some negative event usurps the opportunity underlying the expansion alternative.<sup>2</sup> An important implication of this assumption is that the threshold investment level is not directly affected by the degree of uncertainty surrounding future cash flows. Second, the investment decision cannot be delayed.

-

<sup>&</sup>lt;sup>2</sup> Kenneth Arrow (1968) pp. 8-9) was one of the first to recognize that most investments have some degree of irreversibility associated with them, remarking that "there will be many instances in which the sale of capital goods cannot be accomplished at the same prices as their purchase". The importance of investment irreversibility has infiltrated theories of the firm. It is now recognized that market frictions (irreversibilities) at least partially explain persistent differences in performance across industries and across organizations (Ghemawat, 1991).

An emerging stream of research has shown that the ability to delay irreversible investments, such as those associated with entry, can profoundly affect the decision to invest (Dixit, 1989; Pindyck, 1991). It also undermines the theoretical foundation of neoclassical investment models and invalidates the NPV rule as traditionally taught.<sup>3</sup> When a firm makes an irreversible investment, it exercises its option to invest. It gives up the possibility of waiting for new information to arrive that might affect the desirability or timing of entry. This lost option value is an opportunity cost that must be included as part of the cost of investment. As a result, the NPV rule must be modified such that investment should occur only if NPV is at least equal to the value of keeping the option alive. Investment rules that ignore this opportunity cost (i.e., the value of keeping the option alive) can be grossly in error (McDonald & Siegel, 1986).

Lippman and Rumelt (1985) and Dixit (1989) were the first to place these concerns in the context of market entry. In their models, the firm continuously decides whether it should enter by sinking investment, or wait until the next period and then decide whether to enter. The problem becomes a dynamic tradeoff in which the firm delays entry as long as the opportunity cost of foregoing the option to wait an extra period is higher than the gain from entering in the current period. Based upon this decision rule, factors can then be identified that increase the value of waiting and raises the threshold, or trigger point, at which expansion should occur.

The standard result in option pricing theory is that the value of an option increases with an increase in the volatility in the underlying asset. In the case of market entry, the asset can be viewed as the product market investments used in serving the market. Expected returns from entry will may be uncertain due to forces exogenous to firm control, such as changing demand and supply characteristics, technological changes, or legislation. Greater exogenous uncertainty

<sup>&</sup>lt;sup>3</sup> Myers (1984) provides a complete discussion of the inadequacy of discounted cash flow methods in the presence of real options.

should raise the value of the option to wait and enter in the future increases with uncertainty, the more uncertain the environment, the more likely a firm is to wait before it decides to exercise its option to enter. Thus, as uncertainty increases, the likelihood of entry should decrease given that the costs of entry are irreversible. However, as the reversibility of investments increases, the opportunity costs of entering should also decrease. In this case, option models converge to neoclassical investment models, and uncertainty would have no direct effect on the entry decision (after controlling for systematic risk). Thus, theory dictates that the effect of uncertainty should be *moderated* by the degree of irreversibility pertaining to the investment.

# 2.2 Industry and Firm Factors Affecting Entry Irreversibility

Factors influencing the degree of irreversibility or sunkness may reside at the industry or the firm level of analysis. Across industries, entry may require different *types* or *magnitudes* of investment, both of which affect irreversibility. For example, some industries may necessitate more investment in facilities or equipment. Investments in tangible resources, such as plant and equipment, and intangible resources, such as R&D, may vary in salvage value across time and across industries, particularly given unique exogenous shocks to industries in which firms enter. As a result, firms should be slower to enter industries requiring a higher absolute level of investment in either tangible or intangible resources because more dollars are at risk.

Investments in intangible resources may be especially susceptible to exogenous shocks. Such resources are likely to suffer from market failure (Arrow, 1968) making trade on the open market difficult relative to physical assets (Long & Malitz, 1985).

Firm-specific factors may also affect the firm's ability to re-deploy, or avoid sunk cost investments. For example, previous diversification decisions made by firms will determine the

<sup>4</sup> Our definition of uncertainty corresponds to total variance of an asset price, not purely the systematic component of variance.

types of resources and capabilities already possessed by the firm. If these resources lend themselves to entry into related industries, and the firm has excess capacity, then entry may be less costly and more attractive. More relevant to our theory, if investments in the new industry can be easily re-deployed back to other industries in which the firm already competes, then the entry decision will be viewed as more reversible. Firms targeting industries more unrelated to their portfolio of businesses should have more irreversibility. Thus, the effect of uncertainty should vary depending on how related the target industry is to the firm's existing portfolio.

#### 2.3 The Confounding Effect of Growth Options on Uncertainty

While the real option literature has tended to emphasize the opportunity costs of foregoing the option to wait, it should be noted that the presence of growth options may confound the effect of uncertainty on the likelihood of entry. Entry provides growth options if it gives the firm opportunities to make follow-on investments. Increases in volatility will increase the value of such options and increase the incentives to invest now (Roberts and Weitzman, 1981; Kulatilaka and Perotti, 1998). Any particular investment may have some characteristics increasing the value of defering investment and others that increase the value of the option to grow (Abel, et. al, 1996). Since uncertainty has an inverse effect on the value of these two options, the net effect of uncertainty on investment depends on which of these two options dominate. To our knowledge, there has been no empirical assessment as to whether the existence of growth options moderates the effect of uncertainty. Nevertheless, controlling for the presence of growth options is important when assessing to what extent the value of the option to delay is influenced by the moderating effect of irreversibility on uncertainty.

In summary, of real options theory offers the following propositions on how uncertainty and investment irreversibility jointly influence firm entry decisions:

- 1. The degree of irreversibility of the expansion will moderate the relationship between uncertainty and the likelihood of entry. Specifically, uncertainty will have a stronger negative effect on entry as irreversibility increases.
- 2. Irreversibility may differ across *industry* contexts, especially when industries require greater amounts of investment in either tangible or intangible assets.
- 3. Irreversibility may differ across *firm* contexts, particularly when firms can re-deploy assets to other uses at low marginal costs.

#### 3. Empirical Precedents

Although a significant body of theoretical work has been done in recent years extending option-based models of investment, there remain significant opportunities to empirically examine these models. For example, relatively little empirical research has examined the central proposition that the effect of uncertainty on entry decisions is moderated by the degree of irreversibility. One exception is Campa (1993), who examined the effect of exchange rate uncertainty on entry by foreign firms in the United States, and found that at higher degrees of irreversibility, uncertainty had a larger negative effect on entry. He measured irreversibility in two ways: (1) the ratio of fixed assets to net wealth of all U.S. firms in an industry, and (2) the ratio of media expenditures to company sales by all U.S. firms in each industry. Only the first interaction term significantly explained entry. Given Campa's focus on international expansion, his few (industry-level) measures of irreversibility, his concentration on only one specific type of uncertainty (exchange rate), and his lack of overwhelming support for significant moderating effects, further work is needed to determine whether entry is affected by the relationship between uncertainty and irreversibility in domestic expansion settings.

Both our study and Campa's view entry as a function of two factors: expected returns and entry thresholds. Options directly influence entry thresholds, but since entry thresholds are not observable, options are proposed to influence entry, holding expected returns constant. Our approach differs from that of most other studies, which have focused on the direct effect of uncertainty on *levels of investment*. Underlying this approach is the implicit assumption that all investments are *equally* irreversible. Such an approach compromises validation of the interaction effect between uncertainty and irreversibility.

In addition, most empirical studies have examined the effect of various types of macroeconomic uncertainty on aggregate (total industry or macroeconomic) investment. For example, Pindyck (1986) demonstrates a negative correlation between the variance of lagged stock market returns and aggregate investment spending in the US. Episcopos (1995) found that the level of fixed investment in the U.S. is inversely related to five different types of macroeconomic uncertainty. Carruth, Dickerson, and Henley (2000) summarize various aggregate empirical studies that attempt to correlate investment with a proxy measure of uncertainty. The broad consensus is that there is a significant negative relationship between macroeconomic uncertainty and investment. The interaction between uncertainty and irreversibility has not been explicitly tested in these studies of aggregate investment.

While there are expected to be significant option-based effects on aggregate investment (Caballero & Pindyck, 1996), real options approaches to investment specifically attend to *firm-level* trigger points. Heterogeneity across industries and firms may mask the true effects of uncertainty in aggregate-level studies. For these reasons, Carruth, Dickerson, and Henley (2000) have argued that it is at the firm level of analysis that any measurable impact of irreversibilities

on investment would most likely be detected.<sup>5</sup> However, they note that the results obtained by firm-level studies of investment are far less conclusive regarding the relationship between uncertainty and investment. While a number of studies report the expected negative relationship between uncertainty and investment (Campa, 1993; Guiso & Parigi, 1999; Huizinga, 1993), a number of firm-level studies also report weak or no relationships (Campa & Goldberg, 1995; Driver, Yip, & Dakhil, 1996; Leahy & Whited, 1996). One potential explanation for these contradictory findings at the level of the firm is inattention to firm-level differences in investment threshold levels. Unless firm-level studies control for differences in firm thresholds, the effects of uncertainty may be ambiguous or inconclusive. As argued earlier, firm thresholds may differ because their investments have different degrees of irreversibility.

In many ways, this body of research is related to studies examining the effect of firm diversification on entry (Montgomery et al., 1991; Silverman, 1999). A central prediction of the diversification school is that firms better able to leverage existing assets into expansion alternatives will be more likely to enter. Surprisingly, there has been little integration of real options theory into the diversification literature. Recently, however, a few studies have considered that real options theory has implications for a firm's mode of entry (Kogut, 1991; Folta, 1998). The underlying assumption of these studies is that entry modes differ in their degree of irreversibility. For example, Folta (1998) argued that equity partnerships involve less commitment than complete acquisitions, and found that uncertainty increases the likelihood that firms will expand via equity partnerships versus complete acquisition. Mode of entry is a specific case of our more general focus on entry. Thus, our work is complementary to the mode

\_

<sup>&</sup>lt;sup>5</sup> See their paper for a more thorough discussion of why firm-level analysis offers a number of potential advantages for testing real option theory.

of entry literature, yet provides a more general examination of the entry decision and a more precise test of the interaction between uncertainty and irreversibility.

Collectively, these empirical precedents point toward our expectations, yet differ critically from the present study. Most studies focus on the relationship between uncertainty and investment levels and implicitly assume that all investments are equally irreversible across firms and/or industries. Only Campa (1993) has examined the moderating effect of irreversibility on uncertainty. However, his context was entry into international markets, and he did not consider that irreversibility may differ across firms. Further, most of the studies use industry level, not firm level data, and thus are limited in their ability to address issues relating to firm-specific resource bases, one important focus of this study.

#### 4.0 Research Method

#### 4.1 Data

The primary sources of data for this study come from the Compustat industrial and business segment databases, commonly referred to as the Compustat I & II databases.

Compustat I contains detailed financial information at the level of the firm for all public companies in the United States. Accordingly, most of our firm level variables were drawn from this database. Compustat II provides financial data for each of a firm's business segments from 1980 to the 1999. In Compustat II, firms may report up to ten individual business segments, with six 4-digit SIC codes per segment (2 primary SSIC's, and 4 product PSIC's). Thus, for any given year, a particular firm may report up to sixty SIC codes per year. We use Compustat II to define entry, and to generate most of our industry-level variables. Our analysis included all firms listed in Compustat I and II from 1980 to 1999, which encompassed 17,861 unique firms and 143,770 firm/year observations. Thus, our data represent a large sample of firms operating

in many different industries over a significant period of time. For the purposes of this study, these qualities are advantageous since they should yield significant variation in both industry-level uncertainty and the extent of irreversibility associated with entry decisions.

Despite the potential strengths of this data, potential issues with the use of Compustat must also be considered. Probably the greatest concern is that industry segment SIC codes may be inconsistently assigned, making it difficult to observe legitimate instances of new business activity. This shortcoming in the data is one of the principle reasons we decided to move to a higher level of aggregation than 4-digit SIC.<sup>6</sup> Additionally, there is some discretion in how firms group business activity in a single segment, leading to the potential agglomeration of diverse business activities into the same segment. However, Davis and Duhaime (1992) argue that the problem should not be substantive because of FASB reporting requirements. Furthermore, by moving to a higher level of aggregation, we attenuate this concern.

Following Fama and French (1997) and Brav and Gompers (1998), we reclassified reported SICs into forty-eight possible industries, for which Fama and French (1997) measure industry level stock market returns. Broadly speaking, these 48 industries represent a recoding of the SIC system to somewhere between the 2-digit and 3-digit level. Using the forty-eight industry classification, we identified 12,127 instances of entry between 1981 and 1999 in the Compustat data. Entry was defined as entry by an existing firm into an industry in which the firm had not reported involvement in the previous two years. After classifying each of the reported SICs to our industry definitions, we examined whether any reported SSIC's in the current year matched any of the possible sixty SIC's reported by the firm in the two previous

\_

<sup>&</sup>lt;sup>6</sup> For example, while a segment may be coded as manufacturing disk drives (SIC 3684) one year and computer peripheral parts (SIC 3688) the next, it will probably not be classified outside of the more general category of the "computer industry" (Industry number 35 under our system).

years. If they did not match, and the firm was in Compustat II in the prior year, they were coded as an entry. It should be noted that one firm entering several new industries would provide multiple instances of entry.

#### 4.2 Model Specification

Real options theory predicts that entry will occur when the discounted value of future cash flows, less the opportunity cost associated with foregoing the option to delay entry, exceeds zero. In traditional discounted cash flow models, there are a number of factors that might influence a firm's estimate of the present value of future cash flows. The current rate of industry profitability, expectations regarding future growth, the size of potential entry barriers into the industry will all influence a firm's expectations about the NPV of investment. Similarly, the value of the option to wait will also be affected by a number of different factors all related to the degree of uncertainty and irreversibility associated with the investment. To model the relationship between the likelihood of entry and these factors, we specify the following model. Specificially, we model the probability that firm K enters industry J at time t as:  $P_t(K,J) = f(NPV_t(J,K), U_t(J), I_t(J,K), U_t(J)*I_t(J,K)]$ , where expected  $NPV_t(J,K)$  is a function of both industry J and firm K specific factors,  $U_t(J)$  represents industry-level uncertainty,  $I_t(J,K)$  represents variables affecting the degree of irreversibility for investments in industry J by firm J0, and J1, J2, J3, J4, J5, J5, J5, J6, J7, J8, J8, J8, J9, J9, J9, J1, J1, J2, J3, J4, J5, J5, J5, J5, J5, J6, J7, J8, J8, J8, J9, J1, J1, J1, J2, J3, J4, J5, J5, J5, J5, J5, J5, J5, J7, J8, J8, J8, J8, J9, J9, J9, J1, J1, J2, J3, J4, J5, J5, J5, J5, J5, J5, J5, J5, J7, J8, J8, J9, J1, J1, J1, J2, J3, J4, J5, J5, J5, J5, J5, J5, J5, J5, J7, J8, J8, J9, J9, J9, J9, J9, J1, J1, J2, J3, J4, J5, J5, J5, J5, J5, J5, J5, J5, J5, J7, J8,

# 4.3 Logit Procedure

Similar to Montgomery and Hariharan (1991), we test our model with a series of multivariate binomial logit models that compare 12,127 instances of entry with a random sample

of 60,000 non-entries.<sup>7</sup> The sample of non-entries was created by randomly generating (with replacement) a sample of 60,000 firm-year observations, and randomly assigning industries to these observations in which firms had not competed for the previous two years. The sample of non-entries represents only a fraction of the set of all potential non-entries. Specifically, 127,073 firm-year observations combined with 48 potential industries implies 6,099,504 potential industry-firm-year combinations. Although state-based sampling does not give a pure random sample (the probability that an instance of entry is in the sample is 1, whereas the probability that an instance of non-entry is in the sample is 1.0% [60,000 / (127,073 \* 48 – 143,770)], there are important advantages in state-based sampling. If a purely random sample were selected, it would consist overwhelmingly of instances of non-entry, since instances of entry form such a small proportion of the total population. In comparison to the derived state-based sample, the information yield from a purely random sample would be very small. Hence, estimates from state-based sampling have superior efficiency properties (see Manski and McFadden, 1981).

To be unbiased and consistent, however, the maximum likelihood function derived from state-based samples must be modified. This can be done through a straightforward correction where ln[(proportion of non-entries in the population that are included in the sample)/(proportion of entries in the population that are included in the sample)], or, in this case, ln(0.01/1.00), is added to the constant term. The values of all other parameters are unaffected by this procedure.

## 4.4 Independent Variables

## 4.4.1 Measurement of Uncertainty

\_

<sup>&</sup>lt;sup>7</sup> The data do not distinguish between direct entry and entry by acquisition. Nevertheless, the motives for delaying expansion outlined above should apply to both modes of entry. While it is possible that the scale of entry differs across these modes, we do not expect the degree of irreversibility per dollar invested to differ across these modes.

In any empirical investigation of real options theory, uncertainty is one of the most difficult and important constructs to quantify. One approach that has been commonly employed in studies of economics and strategy is to simply calculate the variance of some output (e.g., stock price, GDP, price, exchange rate) over time. However, for our purposes, this approach has two critical shortcomings. First, it fails to account for the trends in the data. Although trends in the data will increase the measured variance, they may not constitute an element of uncertainty if they are predictable. Thus, our measure of uncertainty should only consider variance about a predicted trend. Second, this approach does not allow for the possibility that the variance may be heteroskedastic (i.e., having-non constant variance over time), a characteristic that is typical of many economic time series (Bollerslev, 1986).

To address both of these concerns, we measure uncertainty with the conditional variance generated from generalized autoregressive conditional heteroskedasticity (GARCH) models, (Bollerslev, 1986). The conditional variance produced by GARCH models provides a time varying estimate of uncertainty about the trend. Fitting this model on monthly data for forty-eight different series of stock market indices from 1950 to 1998, enabled us to approximate unique time-varying estimates of uncertainty for forty-eight different industries. Specifically, we employ the GARCH-in-mean, or GARCH-M model, with a functional form as follows:

(1) 
$$y_t = x_t \beta + \delta \sqrt{h_t} + \varepsilon_t$$

(2) 
$$\varepsilon_t = \sqrt{h_t} e_t$$

-

<sup>&</sup>lt;sup>8</sup> For example, if a time series such as a market index were to increase 10% every year, a rational observer would believe there to be essentially no "uncertainty" in this series. However, a simple variance of these returns would exhibit a positive variance over every time interval.

<sup>&</sup>lt;sup>9</sup> As an alternative measure of uncertainty, we estimated the conditional variance of industry profitability (Net Income / Assets) from quarterly Compustat I data. In using this measure of uncertainty to estimate entry, we produced qualitatively identical results to those reported in table 2.

(3) 
$$h_t = w + \sum_{i=1}^{q} \alpha_i \varepsilon_{t-i}^2 + \sum_{j=1}^{p} \gamma_j h_{t-j}$$

(4) 
$$e_t \sim IN(0,1)$$

The GARCH process is parameterized by two values, p and q in Equation (3). The first value, p, specifies the number of lags for the squared error terms. The second parameter, q, relates to the number of past variances to be included in the computation of the current variance. In general, a one period lag on both parameters (i.e., a GARCH[1,1] model) provides excellent fit for modeling a wide variety of asset prices (Solnik, 1996). Diagnostic checks of our data indicate that the parsimonious GARCH(1,1) model provides excellent fit.<sup>10</sup>

We used the monthly conditional variances ( $h_t$ ) generated from the GARCH-M(1,1) model to develop yearly measures of INDUSTRY UNCERTAINTY by taking the square root of the average conditional variance over the calendar year. We lagged these measures twelve months to account for the fact that average entry for the year occurred in the middle of the year (i.e., during June or July), and there may be a time lag of six months between the entry decision and the actual entry as recorded in Compustat. For example, if entry was observed at the end of 1985, we considered uncertainty from January 1984 to December 1984.

#### 4.4.2 Variables Pertaining to Irreversibility

A number of variables are developed to approximate industry and firm-specific irreversibility. To capture the degree irreversibility in destination industries, we use average industry R&D expenditures (INDUSTRY R&D) and average industry capital expenditures (INDUSTRY CAPEX). These measures were calculated from data in Compustat II by

We evaluated model fit based on verification that the residuals were distributed as white noise; the statistical significance of the hyperparameters (i.e., the  $\omega$ ,  $\alpha_1$ , and  $\gamma_1$  in Eq. 4); and comparison of the model with alternative lag structures.

We also created a measure of uncertainty based on six month lags, but found that the coefficient for uncertainty did not change substantively.

aggregating across each of our forty-eight industries, and dividing by the number of firms competing in each industry. This aggregation method is identical to the one used by Montgomery and Hariharan (1991). We transform average industry R&D by a logarithm to correct for skewness.<sup>12</sup>

The relatedness of a destination industry to the prospective entrant should also have an effect on the degree of investment irreversibility. Traditional measures of industry relatedness focus squarely on the similarity of the industry SICS. However, this approach is problematic because it is unclear whether the same degree of similarity exists between all pairs of SICS at all levels of analysis. In addition, such an approach in this particular study is difficult due to our use of French & Fama's 48 industry classification. To address these issues, we adopt a measure of industry relatedness similar to that proposed by Teece, Rumelt, Dosi, & Winter (1994), which measures the likelihood that a firm operating in industry *j* will also operate in industry *m*, corrected for the expected degree of relatedness under the null hypothesis that diversification is random. Our measure differs from the previous authors in that we calculate the index for the 48-industry classification used in this study and allow the measure to vary over time.

Our measure of unrelatedness (UNRELATED) between any single given industry and a firm's existing portfolio of industries was the minimum distance between that industry and all of the industries that are already in the firm's portfolio. We felt that the minimum distance between the target industry and the firm's existing portfolio was appropriate since relatedness was predicted to impact the redeployability of the assets committed to the new industry (and hence

\_

<sup>&</sup>lt;sup>12</sup> Consistent with previous research (Minton & Schrand, 1999), INDUSTRY R&D was assumed to be zero if no R&D expenditure was listed in Compustat for a given firm/year observation. One was added to the measure prior to the logarithmic transformation.

their degree of "sunkness"). Redeployability should be most affected by the distance between the new industry and their next most appropriate use within the firm.

# 4.4.3 Measurement of Growth Options

Whether the option to delay or the option to grow dominates the entry decision may depend on the value of the industry's market-to-book ratio. As argued by Myers (1977), a high market-to-book ratio may be associated with a higher proportion of growth opportunities relative to assets-in-place. For firms targeting entry into industries with low market-to-book ratios, the option to wait may dominate the timing decision. As the market-to-book increases, the option to grow should take on greater importance, influencing, and possibly changing, the effect uncertainty has on the likelihood of entry. Such an effect is similar to one discussed by Kulatilaka & Perotti (1998). To control for the presence of growth options we use the industry market-to-book ratio. INDUSTRY M/B was calculated from Compustat I by aggregating across firms for the industry designated at the firm's primary line of business.

# 4.4.4 Variables Pertaining to Other Industry and Firm Characteristics Influencing NPV

As argued previously, a firm will enter into an industry when the NPV of expected cashflows, less the opportunity costs of foregoing the option to wait, are greater than zero. Our measures of uncertainty and irreversibility allow us to account for the opportunity costs of the option to wait. However, a model that properly reflects the specification of the theoretical model should also include industry and firm variables correlated with the present value of investment. We first describe our industry-level variables approximating forecasted performance, systematic

<sup>&</sup>lt;sup>13</sup> Following Miller and Modigliani (1961), firm market value is represented by the present value of (earnings generated by) assets-in-place and growth opportunities, which can be approximated by tangible and intangible assets, respectively; whereas book value is the present value of assets-in-place.

risk, and industry structure, and then discuss our firm-level variables meant to approximate the firm's unique ability to enter new industries.

Forecasted industry performance is approximated by predicted industry returns, average firm sales, and industry growth. Predicted returns for the destination industry (INDUSTRY PRETURNS), are taken from the GARCH process identified in equation 1,  $\tilde{y}_t$ . We measure the industry sales (INDUSTRY SALES) from Compustat II by aggregating sales for each year across industries and dividing by the number of firms in each industry. Industry growth in sales (INDUSTRY GRSAL) measures the total growth in industry sales versus two years prior. This measure should control for the possibility that option valuations vary across industry or product life cycles (Bollen, 1999). Destination industries with higher predicted returns, higher average firm sales, and higher growth should be more attractive entry candidates.

The variable BETA provides, for each industry, an index of that industry's systematic risk relative to the whole market and is calculated in five-year increments to account for the possibility that systematic risk changes over time. On the one hand, an increase in systematic risk will tend to increase the discount rate, lowering NPV. At the same time, systematic risk may be associated with higher expected returns, raising NPV. Determining which of these effects will dominate is difficult to ascertain. It is measured by the covariance of monthly returns on each industry portfolio to the entire market portfolio.

We employ a Herfindahl ratio (CONC) to capture industry structure. We expect that more concentrated industries (i.e., higher Herfindahls) will have more significant entry barriers. This index is calculated for each industry by summing the squared market shares of all competitors in that industry. One disadvantage of this measure is that only accounts for publicly traded firms. However, for the vast majority of industries, total industry sales are dominated by

publicly traded firms, minimizing measurement error. Furthermore, we do not understand how this potential problem would bias our results in any way that would favor our hypotheses.

A number of firm-specific variables help us establish which of these large, established firms are the most likely to pursue diversified expansion. FIRM DIVERS is the annual degree of firm diversification, as measured by the sum of squared shares of each of the firm's business segments, where business segments are defined as one of the 48 industries identified earlier. This measure has been commonly applied elsewhere and represents a variety of resources that may exist within a firm. We expect that a broader range of resources will lead to more entry. FIRM CAPI and FIRM R&D represent capital intensity and R&D intensity, respectively, and calculated by dividing firm capital expenditures and R&D expenditures by total firm sales. If firms have a particular competence for managing in capital intensive industries, this know-how may provide a basis for diversified expansion. Further, diversified expansion may be motivated by unused physical productive capacity. Resources represented by R&D intensity are likely to fuel diversified expansion because capacity constraints for these resources tend to be soft and they are likely to suffer from market failure, making trade on the open market difficult. Finally, FIRM SALES is measured by the logarithm of total firm sales.

All variables are calculated annually. Except for INDUSTRY UNCERTAINTY and INDUSTRY PRETURNS, all variables are lagged one year to prevent endogeneity with our dependent variable. Descriptive statistics and correlations are detailed in table 1.

[Insert tables 1 and 2 about here]

#### 5. Results

We tested the significance of the theoretical model by examining whether the addition of independent variables significantly improved the ability to explain the choice between entry and

non-entry. Table 2 reports the results from a model reflecting traditional investment theory (1), and models reflecting the real options approach (2-8). While each of these models include year fixed effects, they are not reported. An overview of the results shows that the general premise of the research, that entry is influenced by uncertainty, irreversibility (firm-specific and industry-specific), and the interaction between uncertainty and irreversibility, receives strong support.

# 5.1 Main Effect of Uncertainty

One key premise of real option theory is that, in the presence of irreversibility, uncertainty should have a direct negative effect on the timing of a firm's entry. Model 2 adds INDUSTRY UNCERTAINTY. A likelihood ratio test comparing model 2 with model 1 revealed a chi-square statistic of 240.0, far exceeding the critical value for one degree of freedom. This suggests that that the addition of industry uncertainty significantly improves model fit (p < 0.0001) relative to the traditional investment model. The negative coefficient (p < 0.0001) for INDUSTRY UNCERTAINTY is consistent with expectations, and suggests that greater industry uncertainty decreases the likelihood of entry.

# 5.2 Moderating Effects to Uncertainty

Model 3 adds the interaction term involving INDUSTRY UNCERTAINTY and INDUSTRY M/B to account for the varying impact of uncertainty in the presence of growth options. A likelihood ratio test comparing model 3 to model 2 revealed a chi-square statistic of 417.3, suggesting the interaction term significantly (p < 0.0001) improved explanatory power. As suggested in previously cited theoretical work, controlling for growth options appears to be important when studying the effects of uncertainty. After controlling for the presence of growth options on the effect of uncertainty, we incrementally add interaction terms (4-7) relating to irreversibility and predicted returns. Separate likelihood ratio tests comparing models 4-7 to

model 3 suggest that all interaction terms significantly (p < 0.0001) improve explanatory power. This finding confirms our main expectation – that uncertainty is moderated by irreversibility.

Model 8 contains the full set of interaction terms. Since the coefficients and standard errors for these interaction terms do not vary considerably across models, we use model 8 to interpret individual coefficients. While the main effect of INDUSTRY UNCERTAINTY is positive in model 8, one cannot interpret total effects in the presence of significant interactions. The positive coefficient for the interaction between INDUSTRY UNCERTAINTY and INDUSTRY M/B is consistent with the view that the effect of negative effect of uncertainty becomes less pronounced in the presence of growth options. In the next section, we discuss this relationship in greater detail.

After controlling for the presence of growth options, we expect that the effect of uncertainty on entry will be moderated by the degree of irreversibility associated with entry. The negative coefficients for the interaction between INDUSTRY UNCERTAINTY and INDUSTRY CAPEX and between INDUSTRY UNCERTAINTY and INDUSTRY R&D are consistent with expectations, and suggest that the negative effect of INDUSTRY UNCERTAINTY is more pronounced when considering entry into industries with higher levels of average capital expenditures (p < 0.0001) and higher levels of average R&D expenditures (p < 0.0001).

The negative coefficient (p < 0.0001) for the interaction between INDUSTRY

UNCERTAINTY and UNRELATEDNESS of target industries is consistent with expectations—
the negative effect of uncertainty is more pronounced when entering industries that are less
related to the firm's existing operations. This finding supports our view that irreversibility
differs across firms and that when combined with uncertainty, these firm-level differences bear
upon entry timing decisions.

While the interaction between INDUSTRY UNCERTAINTY and INDUSTRY
PRETURNS does not directly pertain to theory discussed earlier, standard expectations from real options theory are that we should see more entry when uncertainty is low and predicted returns are high. This is because the option to defer has less value at lower levels of uncertainty. As uncertainty increases so does the value of holding an option, so we should see less entry, even in industries with high predicted returns. The negative coefficient (p < 0.0001) for this interaction term confirms this expectation, adding validity to our option model.

#### 5.3 The Probability of Entry and Its Elasticity with Respect to Uncertainty

The empirical results reported above, in general, suggest that our results are statistically significant and support our proposed hypotheses. However, since we are using a logit estimation method, it is difficult to interpret the economic significance of these variables without converting the estimated log odds to actual probabilities. The results of this conversion are reported in Table 3, which illustrates the how uncertainty influences the probability of entry at different levels of the interacted variables: INDUSTRY M/B, INDUSTRY CAPEX, INDUSTRY R&D, UNRELATED and INDUSTRY PRETURNS. While these probabilities may appear to be quite small, a relevant comparison is the ratio of observed entries over the total probability space, i.e., the number of observed entries / the number of possible entries = 12,127 / 6,099,504 = .19%.

#### [Insert Table 3 about here]

Table 3 reveals significant variation in the effect of uncertainty on the probability of entry over the various combinations of interacted variables. The probability of entry varies most with the relatedness of the target industry (UNRELATED) and industry market-to-book (INDUSTRY M/B). Across all levels of uncertainty, higher predicted returns (INDUSTRY PRETURNS)

increase the probability of entry. Table 3 also demonstrates that, in general, as uncertainty increases, the probability of entry decreases, for some variables quite precipitously.

In addition, Table 3 reports the elasticitiy of entry with repect to uncertainty for the various combinations of variables, i.e. the percentage change in the probability of entry given a percentage change in uncertainty. In general, uncertainty has a negative effect on entry, however the strength of the uncertainty effect varies considerably across the variables. For example, when INDUSTRY UNCERTAINTY is at the 50 percent quantile and INDUSTRY CAPEX is at the 10 percent quantile, the elasticity of entry with respect to uncertainty is -1.08 percent, while it is -3.02 percent at high levels (90<sup>th</sup> quantile) of INDUSTRY CAPEX. When INDUSTRY UNCERTAINTY is at the 50 percent quantile and target industries are unrelated (90<sup>th</sup> quantile), elasticity equals -1.57 percent, while when they are related to existing businesses, uncertainty has a smaller negative effect. In contrast to these findings, the elasticity of uncertainty with respect to INDUSTRY R&D seems to be low (i.e., changes in INDUSTRY R&D have little impact on the effect of uncertainty.)

Finally, Table 3 illuminates our findings regarding the interaction between INDUSTRY UNCERTAINTY and INDUSTRY M/B. At low levels of market-to-book, the elasticity of entry with respect to uncertainty is negative, which is consistent with an option to wait perspective. However, as market-to-book increases, the elasticity of entry gradually becomes less negative until the effect switches sign, suggesting that as uncertainty increases, the likelihood of entry also increases. This non-monotonic effect stand in contrast to the monotonic effects associated with all the other interaction variables. These findings support the view that there is less incentive to delay entry when growth options are larger (Kulatilaka and Perotti, 1998; Abel, et.al., 1997).

They also suggest growth options can dominate the option to wait, leading to a positive effect of uncertainty on entry.

#### 5.4 Main Effects of Other Variables

The results for the other independent variables reported in table 2 are largely consistent with expectations and findings from previous research. Firms are more likely to enter industries with higher predicted returns (p < 0.0001), higher growth in sales (p < 0.0001), and higher industry sales (p < 0.0001). Firms are less likely to enter industries with higher average capital expenditures (p < 0.0001) and higher average R&D expenditures (p < 0.0001), consistent with the view that such industries require greater degrees of irreversibility for effective entry, or have higher entry barriers. The effect of INDUSTRY M/B is significant (p < 0.0001), yet nonmonotonic. At levels of INDUSTRY UNCERTAINTY greater than 5.80 (approximately the mean), INDUSTRY M/B has a positive effect on entry, consistent with a growth option argument. When uncertainty is lower, INDUSTRY M/B has a negative effect on entry. This may suggest that when growth options are inconsequential (because of low uncertainty), entry decisions are dominated by concerns about irreversible investment in intangible assets. In contrast to expectations, industry concentration (CONC) has a positive effect on entry. Firms are more likely to enter industries with higher systematic risk, measured by BETA (p < 0.0001), a finding consistent with Holland, Ott, and Riddiough (2000).

Variables pertaining to relationship-specific and firm-specific effects behave largely as expected. We find that entry is much less likely into unrelated industries (p < 0.0001), a finding consistent with our view that unrelated entry is less reversible and past research examining the importance of existing resource profiles. Firm size, measured by FIRM SALES, has a negative

effect (0.0001) on entry.<sup>15</sup> The breadth of a firm's resource base (DIVERS) is a very strong predictor of entry. More diversified firms are more likely to enter new industries (p < 0.0001).

#### 6.0 Summary and Discussion

In this paper we set out to test existing theory that suggests uncertainty in the face of irreversible investment will affect firm entry decisions. Firm-level empirical studies on the effects of uncertainty are few and offer little consistent support of the implications from real options theory. Furthermore, empirical research at both the aggregate and the firm levels have tended to assume that all investments are equally irreversible. We explicitly test for the effects of uncertainty at different degrees of firm- and industry-level irreversibility.

Our findings suggest that uncertainty has the expected negative relationship with entry decisions. While this expectation has been validated at the aggregate level of analysis, very few studies examine the effect of uncertainty on individual firm commitment decisions. Thus, our study, across a broad range of industries and controlling for a variety of industry and firm characteristics, including year fixed effects, provides compelling evidence that uncertainty affects firm behavior. We should note that our test is a conservative one by the inclusion of other industry specific variables that may affect uncertainty. Indeed, the main and moderating effects of uncertainty are much stronger in models excluding industry level variables. Thus, by including those industry variables we believe our findings to be fairly robust. Moreover, as noted in footnote 10, our results for uncertainty were robust to different measures of uncertainty.

<sup>&</sup>lt;sup>14</sup> The positive effect of INDUSTRY PRETURNS on entry holds for levels of INDUSTRY UNCERTAINTY less than 8.97 (97th percentile).

<sup>&</sup>lt;sup>15</sup> We also introduced control variables pertaining to firm market-to-book ratio, growth in firm sales, and firm profitability. However, their introduction did not substantively effect other relationships, so we exclude them because we lack complete information on these variables.

<sup>&</sup>lt;sup>16</sup> A simple regression, estimating uncertainty with the other industry variables, produced an R-squared of close to fifty percent.

Our study is unique in that it provides strong support for the view that increasing degrees of irreversibility lead to a more pronounced negative effect of uncertainty on entry. Uncertainty has a stronger negative effect on entry into industries requiring more investment in capital expenditures or R&D.

In addition, the expected interaction effect is also very important when considering our firm-level measure of irreversibility. Uncertainty has a more pronounced negative effect on entry into more unrelated industries, where assets are less re-deployable across business units. This study is the first attempt that we know of to model the joint effect of uncertainty and firm-specific irreversibility. This is an exciting finding, and speaks directly to the intersection between industry conditions and firm strategy as they relate to entry. In particular, it suggests that firm entry thresholds may be as enduring as firm strategy, and not purely a temporary phenomenon that fluctuates with industry conditions. It also, however, suggests that firm strategy may have long-term implications for industry structure. In more uncertain industry conditions, related diversifiers may be more likely than unrelated diversifiers to inundate the industry; whereas in less uncertain domains, entry may be more likely to come from both types of firms.

We believe that our study illuminates the tradeoff between the option to wait and the option to grow, both of which the entry decisions may involve. Since the value of both options rises with uncertainty, and the two options have opposing effects on the incentive to invest, the theoretical literature suggests that the effect of uncertainty is ambiguous (Abel et al., 1996). We find that the effect of uncertainty is not ambiguous, and depends on the target industry's proportion of growth options relative to assets-in-place. We use industry market-to-book ratios as a measure of growth options relative to assets in place (Myers, 1977). While this

explanation for the strong and consistent effect for the interaction between industry market-to-book and uncertainty on the likelihood of entry observed in Figure 1. We find that uncertainty has a negative relationship with entry across a large portion of the sample, but the negative effect decreases in importance as firms target industries with larger growth options. Ultimately, when growth options become large enough, uncertainty has a positive effect on entry, an expectation aroused by Kulatilaka and Perotti (1998). Given that theoretical work reconciling the tradeoff between options to wait and growth options is very recent, this study may provide important empirical evidence that such a tradeoff exists. Additional work should examine this tradeoff. Strategic considerations such as competitive preemption (Trigeorgis, 1991; Kulatilaka and Perotti, 1998) or learning (Roberts and Weitzman, 1981) may exacerbate the incentive to exercise growth options.

Finally, there are opportunities to explore whether our findings translate to contexts involving other strategic decisions by firms at all levels, including exit from industries, divestment, and new product introductions. Understanding how uncertainty and irreversibility combine to create unique firm investment thresholds is a promising area of research.

#### 7. References

Abel, A. B., Dixit, A. K., Eberly, J. C., Pindyck, R. S. 1996. Options, the value of capital, and investment. *Quarterly Journal of Economics*, 111(3): 753-777.

Arrow, K. J. 1968. Optimal capital policy with irreversible investment. In J. N. Wolfe (Ed.), Value, Capital and Growth, Essays in Honor of Sir John Hicks. Edinburgh, Scotland: Edinburgh University Press.

- Bollen, N.P.B. 1999. Real options and product life cycles. *Management Science*, 45(5): 670-684.
- Bollerslev, T. 1986. Generalized autoregressive conditional heteroskedasticity. *Journal of Econometrics*, 31: 307-327.
- Brav, A. and P.A. Gompers 1998. Myth or reality? The long-term underperformance of initial public offerings: Evidence from venture and nonventure capital-backed companies, *Journal of Finance*, 52: 1791-1822.
- Caballero, R. J., Pindyck, R. S. 1996. Uncertainty, investment, and industry evolution. *International Economic Review*, 37(3): 641-662.
- Campa, J. M. 1993. Entry By Foreign Firms in the United-States Under Exchange-Rate Uncertainty. *Review of Economics and Statistics*, 75(4): 614-622.
- Campa, J. M., Goldberg, L. S. 1995. Investment in manufacturing, exchange rates and external exposure. *Journal of International Economics*, 38: 297-320.
- Carruth, A., Dickerson, A., Henley, A. 2000. What do we knout about investment under uncertainty? *Journal of Economic Surveys*, 14(2): 119-153.
- Davis, R., Duhaime, I. M. 1992. Diversification, vertical integration, and industry analysis: New perspectives and measurement. *Strategic Management Journal*, 13: 511-524.
- Dixit, A. 1989. Entry and exit decisions under uncertainty. *Journal of Political Economy*, 97(June): 620-638.
- Dixit, A. K., Pindyck, R. S. 1994. *Investment Under Uncertainty*. Princeton, N.J.: Princeton University Press.
- Driver, C., Yip, P., Dakhil, N. 1996. large company capital formation and effects of market share turbulence: micro-data evidence from the PIMS database. *Applied Economics*, 28: 641-651.
- Episcopos, A. 1995. Evidence On the Relationship Between Uncertainty and Irreversible Investment. *Quarterly Review of Economics and Finance*, 35(1): 41-52.
- Fama, E. F., French, K. R. 1997. Industry costs of equity. *Journal of Financial Economics*, 43: 153-193.

- Folta, T. B. 1998. Governance and uncertainty: The trade-off between administrative control and commitment. *Strategic Management Journal*, 19(11): 1007-1028.
- Ghemawat, P. 1991. Commitment. New York: The Free Press.
- Guiso, L., Parigi, G. 1999. Investment and demand uncertainty. *Quarterly Journal of Economics*, 114(1): 185-227.
- Holland, A.S., Ott, S.H., Riddiough, T.J. 2000. The role of uncertainty in investment: An examination of competing investment models using commercial real estate data. *Real Estate Economics*, 28(1): 33-64.
- Huizinga, J. 1993. Inflation uncertainty, relative price uncertainty, and investment in US manufacturing. *Journal of Money, Credit and Banking*, 25: 521-54.
- Kogut, B. 1991. Joint ventures and the option to expand and acquire. *Management Science*, 37(1): 19-33.
- Kulatilaka, N., Perotti, E. C. 1998. Strategic growth options. *Management Science*, 44(8): 1021-1031.
- Leahy, J. V., Whited, T. M. 1996. The effect of uncertainty on investment: Some stylized facts. *Journal of Money Credit and Banking*, 28(1): 64-83.
- Lippman, S.A.; Rumelt, R.P. 1985. Demand uncertainty and investment in industry-specific capital. Unpublished, University of California, Los Angeles, September.
- Long, M., Malitz, I. 1985. The investment-financing nexus: Some empirical evidence. *Midland Corporate Finance Journal*, 3(3): 53-59.
- Manski, C. F., McFadden, D. 1981. Alternative estimators and sample designs for discrete choice analysis. In C. F. Manski, D. McFaddent (Eds.), *Structural analysis of discrete data with econometric applications*. Cambridge, MA: MIT Press.
- McDonald, R., Siegel, D. 1986. The value of waiting to invest. *Quarterly Journal of Economics*, 101: 707-728.
- Miller, M. H., F., M. 1961. Dividend policy, growth and the valuation of shares. *Journal of Business*, 34: 411-433.

- Minton, B. A., Schrand, C. 1999. The impact of cash flow volatility on discretionary investment and the costs of debt and equity financing. *Journal of Financial Economics*, 54: 423-460.
- Montgomery, C. A., Hariharan, S. 1991. Diversified expansion in large established firms. Journal of Economic Behavior and Organization: 71-89.
- Myers, S. C. 1977. Determinants of Corporate Borrowing. *Journal of Financial Economics*, 5: 147-175.
- Myers, S. C. 1984. Finance theory and financial strategy. *Interfaces*, 14(1): 126-137.
- Penrose, E. 1959. The Theory of the Growth of the Firm. London: Basil Blackwell.
- Pindyck, R. S. 1986. Capital risk and models of investment behaviour: Sloan School of Management.
- Pindyck, R. S. 1991. Irreversibility, Uncertainty, and Investment. *Journal of Economic Literature*, 29(September): 1110-1152.
- Roberts, K. and M.L. Weitzman (1981). 'Funding criteria for research, development, and exploration projects'. *Econometrica*, 49, pp. 1261-1288.
- Silverman, B. S. 1999. Technilogical resources and the direction of corporate diversification:

  Toward an integration of the resource-based view and transaction cost economics.

  Management Science, 45(8): 1109-1124.
- Solnik, B. 1996. Chapter 3: Appendix of "International Investments" (3rd Ed.). Addison-Wesley Publishing Company: Reading, MA.
- Teece, D. J., Rumelt, R., Dosi, G., Winter, S. 1994. Understanding Corporate Coherence Theory and Evidence. *Journal of Economic Behavior Organization*, 23(1): 1-30.
- Trigeorgis, L. 1991. Anticipated competitive entry and early preemptive investment in deferrable projects. *Journal of Economics and Business*, 43(2): 143-156.

Table 1: Descriptive Statistics and Pearson Correlation Coefficients

	Mean	S.D.	Min.	Max.	-	8	r	4	'n	Pearson 6	Pearson Correlations 6 7 8	ons 8	6	01	=	2	<u> </u>
1 INDUSTRY PRETURNS	1.223	0.399 -0.261	-0.261	3.225													
2 INDUSTRY SALES	11.146	11.146 1.308 6.742	6.742	13.882	0.136												
3 INDUSTRY GRSAL	0.200		2.692 -0.498	77.491	0.000	0.094											
4 INDUSTRY CAPEX	41.757	41.757 54.005 0.493	0.493	412.465 -0.070	-0.070	0.470 0.054	0.054										
5 INDUSTRY R&D	0.283		0.725 0.000	4.666	0.150	0.122	0.122 -0.019	0.039									
6 INDUSTRY M/B	0.867		0.475 0.077	4.746		-0.112	-0.015	0.146 -0.112 -0.015 -0.100 0.067	0.067								
7 CONC	0.058	0.048	0.048 0.008	0.362	0.052	-0.213	-0.213 -0.022	0.080	0.136	0.136 -0.089							
8 BETA	1.048		0.249 0.048	1.540		-0.058	-0.028	0.201 -0.058 -0.028 -0.290	0.049	0.049 0.008 -0.100	-0.100						
9 UNRELATED	0.927	0.058	0.058 0.611	1.000	-0.015	-0.065	-0.065 -0.045	0.051 -0.024	-0.024	090.0	0.075 -0.111	0.111					
10 FIRM SALES	4.017		2.518 0.000	12.061	-0.018	0.024	-0.012	0.024 -0.012 0.028		0.024	0.005 0.024 -0.005 -0.018 -0.104	0.018	-0.104				
11 FIRM DIVERS	0.153		0.241 0.000	0.896	0.024	0.003		-0.026	-0.013	-0.014	0.011 -0.026 -0.013 -0.014 -0.020	0.025 -0.266	-0.266	0.381			
12 FIRM CAPI	9.643	9.643 471.728 0.000		85352.000 -0.004 -0.004 -0.001 -0.002 -0.003 -0.002 -0.002	-0.004	-0.004	-0.001	-0.002	-0.003	-0.002	-0.002	0.001	0.004	0.004 -0.027 -0.009	-0.009		
13 FIRM R&D	0.640	29.329 0.000	0.000	4672.000	0.002		-0.001	0.008 -0.001 -0.001 -0.003	-0.003		0.006 -0.007	0.004	0.000	0.000 -0.033	-0.013 0.137	0.137	
14 INDUSTRY UNCERTAINTY	5.889	19	1.309 2.978	15.047		0.211 -0.433	0.017	-0.285	0.007	0.115 0.087	0.087	0.387	0.000 -0.018	-0.018	0.024	0.024 0.000 0.001	0.001

n = 62,127

Table 2: Maximum Likelihood Estimates for Entry

INDUSTRY PRETURNS   0,2279	Variable Name	1	2	3	4	5	6	7	8
(0.0285)		0.2719 **	0.4090 **	0.4332 **	0.4278 **	0.4343 **	0.4302 **	1.0752 **	
INDUSTRY GRSAL   0.0005   0.0005   0.0005   0.0005   0.0005   0.0005   0.0005   0.0005   0.0005   0.0005   0.0005   0.0005   0.0005   0.0005   0.00029   0			(0.0305)	(0.0299)	(0.0302)	(0.0302)	(0.0301)	(0.1435)	(0.1455)
INDUSTRY GRSAL   0.0005   0.0005   0.0005   0.0005   0.0005   0.0005   0.0005   0.0005   0.0005   0.0005   0.0005   0.0005   0.0005   0.0005   0.00029   0							0.000.00	0.2670 **	0.2764 **
INDUSTRY GRSAL	INDUSTRY SALES			l .	1				
INDUSTRY CAPEX		(0.0098)	(0.0107)	(0.0108)	(0.0108)	(0.0107)	(0.0107)	(0.0108)	(0.010)
(0.0029)   (0.0029)	INDUSTRY GRSAL	0.0005	0.0050	0.0055	0.0106 *	0.0055	0.0051	0.0050	0.0100 *
INDUSTRY CAPEX	INDESTR'I GREE	8			(0.0029)	(0.0029)	(0.0029)	(0.0029)	(0.0030)
INDUSTRY R&D									0.0060.++
INDUSTRY R&D	INDUSTRY CAPEX			ľ	1			1	
INDUSTRY M/B		(0.0004)	(0.0004)	(0.0004)	(0.0013)	(0.0004)	(0.0004)	(0.0004)	(0.0014)
INDUSTRY MB	INDUCTOV D & D	0.2172 **	0.3054 **	-0.2836 **	-0.2663 **	-0.2372	-0.2842 **	-0.2840 **	-0.1833
INDUSTRY M/B  0 0225	INDUSTRT R&D								(0.1347)
CONC		(0.0201)	(=====,		, ,				
CONC  1.3049*** 1.7386*** 1.7994*** 1.9903*** 1.9903*** 1.7922*** 1.8241*** 1.7322*** 1.9043** (0.2069) (0.2069) (0.2076) (0.2076) (0.2069) (0.2069) (0.2078)  EETA  0.9040*** 1.4246*** 1.4127*** 1.5100*** 1.4142*** 1.4111** 1.3999*** 1.4981** (0.0556) (0.0556)  UNRELATED  1.13611*** 1.13715*** 1.14871** 1.14371** 1.14873** 1.9905** 1.14880** 1.13611** 1.113715** 1.14871** 1.14873** 1.9095** 1.14880** 1.01520) (0.07310) (0.0044) (0.	INDUSTRY M/B	0.0235						1	
BETA		(0.0224)	(0.0222)	(0.0842)	(0.0841)	(0.0847)	(0.0869)	(0.0833)	(0.0850)
BETA	CONC	1 2040 **	1 7296 **	1 7004 **	1 0003 **	1 7922 **	1 8241 **	1.7322 **	1.9143 **
BETA  0.9040 ** 1.4246 ** 1.4127 ** 1.5100 ** 1.4142 ** 1.4111 ** 1.3999 ** 1.4981 ** (0.0556)*  UNRELATED  -11.3611 ** -11.3715 ** -11.4871 ** -11.4871 ** -11.4873 ** -8.9995 ** -11.4880 ** -8.3773 ** (0.1526)* (0.7310)*  FIRM SALES  -0.1380 ** -0.1378 ** -0.1406 ** -0.1415 ** -0.1406 ** -0.1406 ** -0.1406 ** -0.1406 ** -0.1408 ** -0.1405 ** -0.1417 ** (0.0044)* (0.0044)* (0.0044)* (0.0044)* (0.0044)* (0.0044)* (0.0044)* (0.0094)* (0.0096)*  FIRM DIVERS  -2.4556 ** -2.4585 ** -2.4753 ** -2.4854 ** -2.4753 ** -2.4772 ** -2.4744 ** -2.4772 ** -2.4744 ** -2.4772 ** -2.4744 ** -2.4772 ** -2.4772 ** -2.4744 ** -2.4772 ** -2.4772 ** -2.4744 ** -2.4772 ** -2.4772 ** -2.4744 ** -2.4772 ** -2	CONC					1			
UNRELATED		(0.2070)	(0.2000)	(0.20,0)	(=====,	·	, ,		
UNRELATED  -11.3611 ** -(0.1527) -(0.1531) -(0.1526) -(0.1048) -(0.0044) -(0.0097) -(0.0097) -(0.0097) -(0.0097) -(0.0097) -(0.0097) -(0.0097) -(0.0097) -(0.0097) -(0.0097) -(0.0097) -(0.0097) -(0.0097) -(0.0098) -(0	ВЕТА	0.9040 **	1.4246 **	1.4127 **	1.5100 **	1.4142 **			
O.   O.   O.   O.   O.   O.   O.   O.		(0.0474)	(0.0599)	(0.0574)	(0.0566)	(0.0565)	(0.0562)	(0.0559)	(0.0556)
O.   O.   O.   O.   O.   O.   O.   O.			11 2715 **	11 4071 **	11 4201 **	11 /972 **	8 0005 **	_11 4880 **	-8 3773 **
FIRM SALES  -0.1380 ** (0.0044) (0.0047) (0.0037) (0.0397	UNRELATED								
FIRM DIVERS  (0.0044) (0.0037) (0.0397)		(0.1327)	(0.1551)	(0.1320)	(0.1322)	(0.1320)	(01, 100)	(=====)	, ,
FIRM DIVERS  2.4556 ** 2.4585 ** 2.4753 ** 2.4854 ** 2.4753 ** 2.4772 ** 2.4744 ** 2.4876 ** (0.0397)  FIRM CAPI  0.0001	FIRM SALES	-0.1380 **	-0.1378 **	-0.1406 **	-0.1415 **	-0.1406 **	-0.1408 **	-0.1405 **	l '
FIRM CAPI		(0.0044)	(0.0044)	(0.0024)	(0.0044)	(0.0044)	(0.0044)	(0.0044)	(0.0044)
FIRM CAPI				0.4550.++	2 4054 **	2 4752 **	2 4772 **	2 4744 **	2 4876 **
FIRM CAPI  0.0001 0.0000 0.0000 0.0000 0.0000 0.0000 0.0000 0.0000 0.0000 0.0000 0.0000 0.0000 0.0000 0.0000 0.0000 0.0000 0.00000 0.0000 0.0000 0.0000 0.0000 0.0000 0.0000 0.0000 0.0000 0.00000 0.0000 0.0000 0.0000 0.0000 0.0000 0.0000 0.0000 0.0000 0.00000 0.	FIRM DIVERS					1			i l
FIRM R&D  (0.0000) (0		(0.0393)	(0.0390)	(0.0377)	(0.0357)	(0.0371)	(0.05)//	(0.001.)	,
FIRM R&D  (0.0000) (0	FIRM CAPI	0.0001	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000
10,0004   (0,0004)   (0,0005)		(0.0000)	(0.0000)	(0.0000)	(0.0000)	(0.0000)	(0.0000)	(0.0000)	(0.0000)
10,0004   (0,0004)   (0,0005)				0.0004	0.0006	0.0004	0.0006	0.0006	0.0006
INDUSTRY UNCERTAINTY  -0.1979 *** (0.0139)  -0.1979 *** (0.0212)  -0.264 † (0.0226)  -0.226)  -0.1077 ** (0.0217)  -0.1406 *** (0.0401)  -0.3532 *** (0.0401)  -0.1406 ** (0.0401)  -0.173)  -0.268 ** (0.0142)  -0.0142)  -0.0142)  -0.0142)  -0.0142)  -0.0142)  -0.0142)  -0.0142)  -0.0142)  -0.0142)  -0.0030 *** (0.00217)  -0.0077 * (0.0217)  -0.0030 *** (0.0217)  -0.4210 * (0.0226)  -0.0229 *** (0.003)  -0.0036 *** (0.0227)  -0.0036 *** (0.0227)  -0.1068 *** (0.0226)  -0.1068 *** (0.0226)  -0.1068 *** (0.0226)  -0.1068 *** (0.0226)  -0.0035 *** (0.0227)  -0.1068 *** (0.0235)  -0.1068 *** (0.0240)  INTERCEPT (adjusted)  -0.5604 ** (0.1291)  -0.2410 * (0.0235)  -0.1068 ** (0.0240)  -0.1461 ** (0.0240)  -0.1461 ** (0.0240)  -0.1461 ** (0.0240)  -0.1461 ** (0.0240)	FIRM R&D	8						1	
INDUSTRY ONCERTAINTY  (0.0139) (0.0212) (0.0226) (0.0212) (0.0226) (0.0212) (0.01107) (0.0401) (0.0401) (0.1173)  (0.01173)  INDUSTRY M/B x INDUSTRY UNCERTAINTY  (0.0142)  (0.0142)  (0.0142)  (0.0142)  (0.0142)  (0.0142)  (0.0142)  (0.0142)  (0.0142)  (0.0142)  (0.0142)  (0.0143)  (0.0143)  (0.0143)  (0.0143)  (0.0003)  (0.0		(0.0004)	(0.0004)	(0.0003)	(0.0003)	(0.0003)	(0.0003)	(0.0003)	(0.000)
(0.0139)	INDUSTRY UNCERTAINTY		-0.1979 **	-0.5110 **	-0.4217 **	-0.5103 **	-0.1406 **	-0.3532 **	0.2654 †
INDUSTRY UNCERTAINTY   (0.0142)   (0.0142)   (0.0148)   (0.0140)   (0.0143)   (0.0143)   (0.0143)   (0.0142)   (0.0142)   (0.0148)   (0.0140)   (0.0143)   (0.0035 ** (0.0003)   (0.0035 ** (0.0003)   (0.0037 ** (0.0003)   (0.0037 ** (0.0003)   (0.0037 ** (0.0003)   (0.0037 ** (0.0003)   (0.0037 ** (0.0003)   (0.0037 ** (0.0003)   (0.0037 ** (0.0003)   (0.0037 ** (0.0003)   (0.0037 ** (0.0003)   (0.0003)			(0.0139)	(0.0212)	(0.0226)	(0.0212)	(0.1107)	(0.0401)	(0.1173)
INDUSTRY UNCERTAINTY   (0.0142)   (0.0142)   (0.0148)   (0.0140)   (0.0143)   (0.0143)   (0.0143)   (0.0142)   (0.0142)   (0.0142)   (0.0148)   (0.0140)   (0.0143)   (0.0035 ** (0.0003)   (0.0035 ** (0.0003)   (0.0037 ** (0.0003)   (0.0037 ** (0.0003)   (0.0037 ** (0.0003)   (0.0037 ** (0.0003)   (0.0037 ** (0.0003)   (0.0037 ** (0.0003)   (0.0037 ** (0.0003)   (0.0037 ** (0.0003)   (0.0003)							0.00(0.00	0 2110 **	0.2000 **
INDUSTRY CAPEX x INDUSTRY UNCERTAINTY  INDUSTRY R&D x INDUSTRY UNCERTAINTY  UNRELATED x INDUSTRY UNCERTAINTY  UNRELATED x INDUSTRY UNCERTAINTY  INDUSTRY PRETURNS x INDUSTRY UNCERTAINTY  INTERCEPT (adjusted)  7.5604 ** 8.8844 ** 10.6744 ** 6.0692 ** (0.0239)  7.5604 ** (0.2191)  7.5604 ** (0.2191)  7.5604 ** (0.2387)  7.5604 ** (0.2452)  7.5604 ** (0.0239)  7.5604 ** (0.3163)  7.75604 ** (0.3163)  7.75604 ** (0.3163)  7.75604 ** (0.3163)  7.75604 ** (0.3163)  7.75604 ** (0.3163)	INDUSTRY M/B x							l	
INDUSTRY UNCERTAINTY  INDUSTRY R&D x INDUSTRY UNCERTAINTY  UNRELATED x INDUSTRY UNCERTAINTY  UNRELATED x INDUSTRY UNCERTAINTY  INDUSTRY UNCERTAINTY  INDUSTRY UNCERTAINTY  INDUSTRY UNCERTAINTY  INDUSTRY UNCERTAINTY  INDUSTRY UNCERTAINTY  INTERCEPT (adjusted)  7.5604 ** (0.2191)  7.5604 ** (0.2191)  7.5604 ** (0.2387)  7.5604 ** (0.2387)  7.5604 ** (0.2452)  7.5604 ** (0.0239)  7.5604 ** (0.3163)  7.5604 ** (0.2191)  7.5604 ** (0.2191)  7.5604 ** (0.2387)  7.5604 ** (0.2452)  7.5604 ** (0.239)  7.5604 ** (0.3163)  7.5604 ** (0.2191)  7.5604 ** (0.2191)  7.5604 ** (0.2387)  7.5604 ** (0.2452)  7.5604 ** (0.239)  7.5604 ** (0.3163)  7.5604 ** (0.2191)  7.5604 ** (0.2191)  7.5604 ** (0.2387)  7.5604 ** (0.2452)  7.5604 ** (0.239)  7.5604 ** (0.3163)  7.5604 ** (0.2191)  7.5604 ** (0.2191)  7.5604 ** (0.2387)  7.5604 ** (0.2452)  7.5604 ** (0.239)  7.5604 ** (0.3163)  7.5604 ** (0.2121)	INDUSTRY UNCERTAINTY			(0.0142)	(0.0142)	(0.0142)	(0.0148)	(0.0140)	(0.0143)
INDUSTRY UNCERTAINTY  INDUSTRY R&D x INDUSTRY UNCERTAINTY  UNRELATED x INDUSTRY UNCERTAINTY  INDUSTRY UNCERTAINTY  INDUSTRY PRETURNS x INDUSTRY UNCERTAINTY  INTERCEPT (adjusted)  7.5604 ** (0.2191)  7.5604 ** (0.2387)  7.5604 ** (0.2452)  7.5604 ** (0.0239)  7.5604 ** (0.3163)  7.5604 ** (0.2191)  7.5604 ** (0.2191)  7.5604 ** (0.2387)  7.5604 ** (0.2452)  7.5604 ** (0.0239)  7.5604 ** (0.3163)  7.5604 ** (0.2191)  7.5604 ** (0.2191)  7.5604 ** (0.2387)  7.5604 ** (0.2452)  7.5604 ** (0.239)  7.5604 ** (0.3163)  7.5604 ** (0.2191)  7.5604 ** (0.2191)  7.5604 ** (0.2387)  7.5604 ** (0.2452)  7.5604 ** (0.239)  7.5604 ** (0.3163)  7.5604 ** (0.2191)  7.5604 ** (0.2191)  7.5604 ** (0.2387)  7.5604 ** (0.2452)  7.5604 ** (0.239)  7.5604 ** (0.239)  7.5604 ** (0.239)  7.5604 ** (0.239)  7.5604 ** (0.239)  7.5604 ** (0.2387)  7.5604 ** (0.2387)	INDUSTRY CAPEX x				-0.0030 **				-0.0035 **
INDUSTRY R&D x INDUSTRY UNCERTAINTY  UNRELATED x INDUSTRY UNCERTAINTY  INDUSTRY PRETURNS x INDUSTRY UNCERTAINTY  INTERCEPT (adjusted)  7.5604 ** (0.2191)					(0.0003)				(0.0003)
INDUSTRY UNCERTAINTY  UNRELATED x INDUSTRY UNCERTAINTY  INDUSTRY PRETURNS x INDUSTRY UNCERTAINTY  INTERCEPT (adjusted)  7.5604 ** 8.8844 ** 10.6744 ** (0.2387)  (0.0217)  (0.0217)  (0.0217)  -0.4210 * (0.1226)  -0.1068 ** (0.1226)  -0.1068 ** (0.0240)  -0.1461 ** (0.0235)  (0.0240)  INTERCEPT (adjusted)  7.5604 ** 8.8844 ** 10.6744 ** (0.2387)  (0.2387)  -0.1068 ** (0.0239)  -0.1068 ** (0.0245)  (0.0239)  -0.1068 ** (0.0245)  -0.1068 ** (0.0240)  -0.1068 ** (0.0240)  -0.1068 ** (0.0240)  -0.1068 ** (0.0240)									0.0101
UNRELATED x INDUSTRY UNCERTAINTY  INDUSTRY PRETURNS x INDUSTRY UNCERTAINTY  INTERCEPT (adjusted)  7.5604 ** (0.2191)  7.5604 ** (0.2387)  7.5604 ** (0.2452)  7.5604 ** (0.239)  7.5604 ** (0.3163)  7.5604 ** (0.2191)	INDUSTRY R&D x								
INDUSTRY UNCERTAINTY  INDUSTRY PRETURNS x INDUSTRY UNCERTAINTY  INTERCEPT (adjusted)  7.5604 ** 8.8844 ** 10.6744 ** 6.0692 ** 10.6718 ** 8.4737 ** 9.7365 ** (0.0240)  INTERCEPT (adjusted)  7.5604 ** 8.8844 ** 10.6744 ** (0.2387) (0.2452) (0.0239) (0.6873) (0.3163) (0.7212)	INDUSTRY UNCERTAINTY					(0.0217)			(0.0222)
INDUSTRY UNCERTAINTY  INDUSTRY PRETURNS x INDUSTRY UNCERTAINTY  INTERCEPT (adjusted)  7.5604 ** 8.8844 ** 10.6744 ** 6.0692 ** 10.6718 ** 8.4737 ** 9.7365 ** (0.0240)  INTERCEPT (adjusted)  7.5604 ** 8.8844 ** 10.6744 ** (0.2387) (0.2452) (0.0239) (0.6873) (0.3163) (0.7212)	INDELATED						-0.4210 *		-0.5218 **
INDUSTRY PRETURNS x INDUSTRY UNCERTAINTY  INTERCEPT (adjusted)  7.5604 ** 8.8844 ** 10.6744 ** 6.0692 ** 10.6718 ** 8.4737 ** 9.7365 ** 5.9772 † (0.2387)  (0.2191)  (0.2387)  (0.2452)  (0.0239)  (0.6873)  (0.3163)  (0.3163)  (0.7212)	l ·								
INDUSTRY UNCERTAINTY  INTERCEPT (adjusted)  7.5604 ** 8.8844 ** 10.6744 ** 6.0692 ** 10.6718 ** 8.4737 ** 9.7365 ** 5.9772 † (0.2191)  (0.0235)  (0.0240)  (0.0235)  (0.0240)  (0.0237)  (0.0237)  (0.0240)	INDUSTRI UNCERTAINTI								
INTERCEPT (adjusted)  7.5604 ** (0.1969)  7.5604 ** (0.2191)  7.5604 ** (0.2387)  7.5604 ** (0.2452)  7.5604 ** (0.0239)  7.5604 ** (0.0235)  7.5604 ** (0.0235)  7.5604 ** (0.0235)  7.5604 ** (0.0240)  7.5604 ** (0.0237)  7.5604 ** (0.02387)	INDUSTRY PRETURNS x								ļ.
(0.1969) (0.2191) (0.2387) (0.2452) (0.0239) (0.6873) (0.3163) (0.7212)	INDUSTRY UNCERTAINTY							(0.0235)	(0.0240)
(0.1969) (0.2191) (0.2387) (0.2452) (0.0239) (0.6873) (0.3163) (0.7212)				10 (511 +	( 0(02 **	10 6710 **	0 4727 **	0.7265 **	5 0772 +
(0.1707) (0.2717) (0.2507) (0.2	INTERCEPT (adjusted)			i		ł .	1	i .	
Likelihood Ratio -22762.63 -22655.29 -22459.86 -22409.04 -22459.80 -22453.94 -22447.75 -22378.84		(0.1309)	(0.2191)	(0.2301)	(0.2732)	(0.0237)	(0.0075)	(3.2.33)	
Likelihood Ratio -22762.63 -22655.29 -22459.86 -22409.04 -22459.80 -22453.94 -22447.75 -22378.84									
	Likelihood Ratio	-22762.63	-22655.29	-22459.86	-22409.04	-22459.80	-22453.94	-22447.75	-22378.84

<sup>†</sup> p < 0.05; \* p < 0.01; \*\* p < 0.0001

Table 3: The Probality of Entry and Its Elasticity with Respect to INDUSTRY UNCERTAINTY at Different Quantiles of INDUSTRY M/B, INDUSTRY CAPEX, INDUSTRY RD, UNRELATED, and INDUSTRY PRETURNS

			INDUSTRY	STRY					INDUSTRY	STRY
	INDUSTRY M/B	RY M/B	CAPEX	ЭΕХ	INDUST	INDUSTRY R&D	UNRELATED	ATED	<b>PRETURNS</b>	URNS
	Prob. Of		Prob. Of		Prob. Of		Prob. Of		Prob. Of	
Quantile	Entry	Elasticity	Entry	Elasticity	Entry	Elasticity	Entry	Elasticity	Entry	Elasticity
Uncertainty at 10% Quantile										
1%	0.284%	-1.50%	0.188%	-0.58%	0.170%	-0.84%	1.371%	-0.47%	0.068%	-0.31%
10%	0.231%	-1.23%	0.186%	-0.62%	0.170%	-0.84%	0.461%	-0.66%	0.109%	-0.59%
20%	50% 0.170%	-0.84%	0.170%	-0.84%	0.170%	-0.84%	0.169%	-0.84%	0.167%	-0.84%
%06	0.106%	-0.23%	0.118%	-1.73%	0.140%	-0.87%	0.117%	-0.90%	0.241%	-1.05%
%66	0.042%	0.97%	0.054%	-3.65%	0.073%	-1.00%	0.105%	-0.92%	0.321%	-1.22%
Uncertainty at 50% Quantile										
1%	1% 0.093%	-2.62%	0.122%	-1.02%	0.091%	-1.46%	0.962%	-0.83%	0.054%	-0.54%
10%	0.092%	-2.16%	0.117%	-1.08%	0.091%	-1.46%	0.281%	-1.16%	0.070%	-1.03%
20%	0.091%	-1.46%	0.091%	-1.46%	0.091%	-1.46%	0.090%	-1.46%	0.089%	-1.46%
%06	0.089%	-0.40%	0.033%	-3.02%	0.073%	-1.53%	0.060%	-1.57%	0.110%	-1.84%
%66	0.086%	1.70%	0.004%	-6.38%	0.035%	-1.75%	0.053%	-1.60%	0.129%	-2.14%
Uncertainty at 90% Quantile										
1%	1% 0.041%	-3.44%	0.088%	-1.34%	0.058%	-1.92%	0.742%	-1.09%	0.045%	-0.70%
10%	0.047%	-2.83%	0.083%	-1.42%	0.058%	-1.92%	0.195%	-1.53%	0.051%	-1.35%
20%	50% 0.058%	-1.92%	0.058%	-1.92%	0.058%	-1.92%	0.057%	-1.92%	0.057%	-1.92%
%06	90% 0.079%	-0.53%	0.013%	-3.96%	0.046%	-2.00%	0.037%	-2.06%	0.062%	-2.42%
%66	99% 0.146%	2.22%	0.000%	-8.37%	0.020%	-2.29%	0.032%	-2.10%	0.066%	-2.80%

<sup>\*</sup> Quantiles represent the location within the range of the variable.

<sup>\*\*</sup> Base rate of entry at median levels of all variables = 0.090%.

<sup>\*\*\*</sup> All calculations are at the medians of other variables in the model.

# ADDITIONAL INSTITUTE PAPERS AVAILABLE FROM THE KRANNERT GRADUATE SCHOOL OF MANAGEMENT

-1995-

	1993
1069	Sugato Chakravarty and John J. McConnell, AN ANAYLSIS OF PRICES, BID/ASK SPREADS, AND BID AND ASK DEPTH SURROUNDING IVAN BOESKY'S ILLEGAL TRADING IN CARNATION'S STOCK.
1070	John J. McConnell and Henri Servaes, EQUITY OWENERSHIP AND THE TWO FACES OF DEBT.
1071	Kenneth J. Matheny, REAL EFFECTS OF MONETARY POLICY IN A 'NEOCLASSICAL' MODEL: THE CASE OF INTEREST RATE TARGETING.
1072	Julie Hunsaker and Dan Kovenock, THE PATTERN OF EXIT FROM DECLINING INDUSTRIES.
1073	Kessan Joseph, Manohar U. Kalwani, THE IMPACT OF ENVIRONMENTAL UNCERTAINTY ON THE DESIGN OF SALESFORCE COMPENSATION PLANS.
1074	K. Tomak, A NOTE ON THE GOLDFELD QUANDT TEST
1075	Alok R. Chaturvedi, SIMDS: A SIMULATION ENVIRONMENT FOR THE DESIGN OF DISTRIBUTED DATABASE SYSTEMS
1076	Dan Kovenock and Suddhasatwa Roy, FREE RIDING IN NON-COOPERATIVE ENTRY DETERRENCE WITH DIFFERENTIATED PRODUCTS
1077	Kenneth Matheny, THE MACROECONOMICS OF SELF-FULFILLING PROPHECIES
1078	Paul Alsemgeest, Charles Noussair and Mark Olson, EXPERIMENTAL COMPARISONS OF AUCTIONS UNDER SINGLE-AND MULTI-UNIT DEMAND
1079	Dan Kovenock, Casper D de Vries, FIAT EXCHANGE IN FINITE ECONOMIES
1080	Dan Kovenock, Suddhasatwa Roy, DYNAMIC CAPACITY CHOICE IN A BERTRAND-EDGEWORTH FRAMEWORK
1081	Burak Kazaz, Canan Sepil, PROJECT SCHEDULING WITH DISCOUNTED CASH FLOWS AND PROGRESS PAYMENTS
	-1996-
1082	Murat Koksalan, Oya Rizi, A VISUAL INTRACTIVE APPROACH FOR MULTIPLE CRITERIA DECISION MAKING WITH MONOTONE UTILITY FUNCTIONS
1083	Janet S. Netz, John D. Haveman, ALL IN THE FAMILY: FAMILY, INCOME, AND

1084	Keith V. Smith, ASSET ALLOCATION AND INVESTMENT HORIZON
1085	Arnold C. Cooper and Catherine M. Daily, ENTREPRENEURIAL TEAMS
1086	Alok R. Chaturvedi and Samir Gupta, SCHEDULING OF TRANSACTIONS IN A REAL-TIME DISTRIBUTED TRANSACTION PROCESSING SYSTEMS: SCALEABILITY AND NETWORKING ISSUES
1087	Gordon P. Wright, N. Dan Worobetz, Myong Kang, Radha V. Mookerjee and Radha Chandrasekharan, OR/SM: A PROTOTYPE INTEGRATED MODELING ENVIRONMENT BASED ON STRUCTURED MODELING
1088	Myong Kang, Gordon P. Wright, Radha Chandrasekharan, Radha Mookerjee and N. Dan Worobetz, THE DESIGN AND IMPLEMENTATION OF OR/SM: A PROTOTYPE INTEGRATED MODELING ENVIRONMENT
1089	Thomas H. Brush and Philip Bromiley, WHAT DOES A SMALL CORPORATE EFFECT MEAN? A VARIANCE COMPONENTS SIMULATION OF CORPORATE AND BUSINESS EFFECTS
1090	Kenneth J. Matheny, NON-NEUTRAL RESPONSES TO MONEY SUPPLY SHOCKS WHEN CONSUMPTION AND LEISURE ARE PARETO SUBSTITUTES
1091	Kenneth J. Matheny, MONEY, HUMAN CAPITAL, AND BUSINESS CYCLES: A MODERN PHILLIPS CURVE-STYLE TRADEOFF
1092	Kenneth J. Matheny, OUTPUT TARGETING AND AN ARGUMENT FOR STABILIZATION POLICIES
1093	Kenneth J. Matheny, THE RELEVANCE OF OPEN MARKET OPERATIONS AS A MONETARY POLICY TOOL
	-1997-
1094	James C. Moore, William Novshek and Peter Lee U, ON THE VOLUNTARY PROVISION OF PUBLIC GOODS
1095	Michael R. Baye, Dan Kovenock and Casper G. deVries, THE INCIDENCE OF OVERDISSIPATION IN RENT-SEEKING CONTESTS
1096	William Novshek and Lynda Thoman, CAPACITY CHOICE AND DUOPOLY INCENTIVES FOR INFORMATION SHARING
1097	Vidyanand Choudhary, Kerem Tomak and Alok Chaturvedi, ECONOMIC BENEFITS OF RENTING SOFTWARE
1098	Jeongwen Chiang and William T. Robinson, DO MARKET PIONEERS MAINTAIN THEIR INNOVATIVE SPARK OVER TIME?

1099	Glenn Hueckel, LABOR COMMAND IN THE WEALTH OF NATIONS: A SEARCH FOR "SYSTEM"
1100	Glenn Hueckel, SMITH'S UNIFORM "TOIL AND TROUBLE": A "VAIN SUBTLETY"?
1101	Thomas H. Brush and Philip Bromiley, WHAT DOES A SMALL CORPORATE EFFECT MEAN? A VARIANCE COMPONENTS SIMULATION OF CORPORATE AND BUSINESS EFFECTS
1102	Thomas Brush, Catherine Maritan and Aneel Karnani, MANAGING A NETWORK OF PLANTS WITHIN MULTINATIONAL FIRMS
1103	Sam Hariharan and Thomas H. Brush, RESOURCES AND THE SCALE OF ENTRY CHOICE: THE COMPETITIVE ADVANTAGE OF ESTABLISHED FIRMS?
1104	Thomas H. Brush, Philip Bromiley and Margaretha Hendrickx, THE RELATIVE INFLUENCE OF INDUSTRY AND CORPORATION ON BUSINESS SEGMENT PERFORMANCE: AN ALTERNATIVE ESTIMATE
1105	Thomas Brush, Catherine Maritan and Aneel Karnani, PLANT ROLES IN THE MANAGEMENT OF MULTINATIONAL MANUFACTURING FIRMS
1106	Thomas H. Brush, Catherine Maritan and Aneel Karnani, THE PLANT LOCATION DECISION IN MULTINATIONAL MANUFACTURING FIRMS: AN EMPIRICAL ANALYSIS OF INTERNATIONAL BUSINESS AND MANUFACTURING STRATEGY PERSPECTIVES
1107	Piyush Kumar, Manohar U. Kalwani and Maqbool Dada, THE IMPACT OF WAITING TIME GUARANTEES ON CUSTOMERS' WAITING EXPERIENCES
1108	Thomas H. Brush, Philip Bromiley and Margaretha Hendrickx, THE FREE CASH FLOW HYPOTHESIS FOR SALES GROWTH AND FIRM PERFORMANCE
1109	Keith V. Smith, PORTFOLIO ANALYSIS OF BROKERAGE FIRM RECOMMENDATIONS
	- 1998 -
1110	Charles Noussair, Kenneth Matheny, and Mark Olson, AN EXPERIMENTAL STUDY OF DECISIONS IN DYNAMIC OPTIMIZATION PROBLEMS
1111	Jerry G. Thursby and Sukanya Kemp, AN ANALYSIS OF PRODUCTIVE EFFICIENCY OF UNIVERSITY COMMERCIALIZATION ACTIVITIES
1112	John J. McConnell and Sunil Wahal, DO INSTITUTIONAL INVESTORS EXACERBATE MANAGERIAL MYOPIA?

1113	John J. McConnell, Mehmet Ozbilgin and Sunil Wahal, SPINOFFS, EX ANTE
1114	Sugato Chakravarty and John J. McConnell, DOES INSIDER TRADING REALLY MOVE STOCK PRICES?
1115	William T. Robinson and Sungwook Min, IS THE FIRST TO MARKET THE FIRST TO FAIL?: EMPIRICAL EVIDENCE FOR MANUFACTURING BUSINESSES
1116	Margaretha Hendrickx, WHAT CAN MANAGEMENT RESEARCHERS LEARN FROM DONALD CAMPBELL, THE PHILOSOPHER? AN EXERCISE IN PHILOSOPHICAL HERMENEUTICS
1117	Thomas H. Brush, Philip Bromiley and Margaretha Hendrickx, THE FREE CASH FLOW HYPOTHESIS FOR SALES GROWTH AND FIRM PERFORMANCE
1118	Thomas H. Brush, Constance R. James and Philip Bromiley, COMPARING ALTERNATIVE METHODS TO ESTIMATE CORPORATE AND INDUSTRY EFFECTS
1119	Charles Noussair, Stéphane Robin and Bernard Ruffieux, BUBBLES AND ANTI- CRASHES IN LABORATORY ASSET MARKETS WITH CONSTANT FUNDAMENTAL VALUES
1120	Vivian Lei, Charles N. Noussair and Charles R. Plott, NON-SPECULATIVE BUBBLES IN EXPERIMENTAL ASSET MARKETS: LACK OF COMMON KNOWLEDGE OF RATIONALITY VS. ACTUAL IRRATIONALITY
	-1999-
1121	Kent D. Miller and Timothy B. Folta, ENTRY TIMING AND OPTION VALUE
1122	Glenn Hueckel, THE LABOR "EMBODIED" IN SMITH'S LABOR-COMMANDED MEASURE: A "RATIONALLY RECONSTRUCTED" LEGEND
1123	Timothy B. Folta and David A. Foote, TEMPORARY EMPLOYEES AS REAL OPTIONS
1124	Gabriele Camera, DIRTY MONEY
1125	Wilfred Amaldoss, Robert J. Meyer, Jagmohan S. Raju, and Amnon Rapoport, COLLABORATING TO COMPETE: A GAME-THEORETIC MODEL AND EXPERIMENTAL INVESTIGATION OF THE EFFECT OF PROFIT-SHARING ARRANGEMENT AND TYPE OF ALLIANCE ON RESOURCE-COMMITMENT DECISIONS
1126	Wilfred Amaldoss, Robert J. Meyer, Jagmohan S. Raju, and Amnon Rapoport,

APPENDICES FOR COLLABORATING TO COMPETE: A GAME-THEORETIC MODEL AND EXPERIMENTAL INVESTIGATION OF THE EFFECT OF PROFITSHARING ARRANGEMENT AND TYPE OF ALLIANCE ON RESOURCE-COMMITMENT DECISIONS

-2000-

1127	Sugato Chakravarty and Kai Li, AN ANALYSIS OF OWN ACCOUNT TRADING BY DUAL TRADERS IN FUTURES MARKETS: A BAYESIAN APPROACH
1128	Sugato Chakravarty, STEALTH TRADING: THE NEXT GENERATION
1129	S.G. Badrinath and Sugato Chakravarty, ARE ANALYST RECOMMENDATIONS INFORMATIVE?
1130	Sugato Chakravarty and Asani Sarkar, THE DETERMINANTS OF LIQUIDITY IN U.S. CORPORATE, MUNICIPAL AND TREASURY BOND MARKETS
1131	Vivian Lei and Charles Noussair, AN EXPERIMENTAL TEST OF AN OPTIMAL GROWTH MODEL
1132	Paul Healy and Charles Noussair, BIDDING BEHAVIOR IN THE PRICE IS RIGHT GAME: AN EXPERIMENTAL STUDY
1133	Kent D. Miller and Zur Shapira, BEHAVIORAL OPTION THEORY: FOUNDATIONS AND EVIDENCE
1134	Kent D. Miller, KNOWLEDGE INVENTORIES AND MANAGERIAL MYOPIA
1135	Gabriele Camera, Charles Noussair, and Steven Tucker, RATE-OF-RETURN DOMINANCE AND EFFICIENCY IN AN EXPERIMENTAL ECONOMY
1136	Timothy B. Folta, Jay J. Janney, SIGNALING FOR RESOURCE ACQUISITION: PRIVATE EQUITY PLACEMENTS BY TECHNOLOGY FIRMS
1137	Michael R. Baye, Dan Kovenock, Casper G. de Vries, COMPARATIVE ANALYSIS OF LITIGATION SYSTEMS: AN AUCTION-THEORETIC APPROACH
1138	Sugato Chakravarty, Asani Sarkar, DO DIFFERENCES IN TRANSPARENCY AFFECT TRADING COSTS? EVIDENCE FROM U.S. CORPORATE, MUNICIPAL AND TREASURY BOND MARKETS
1139	Charles Noussair, Stephane Robin, Bernard Ruffieux, GENETICALLY MODIFIED ORGANISMS IN THE FOOD SUPPLY: PUBLIC OPINION VS CONSUMER BEHAVIOR

Gabriele Camera, SEARCH, DEALERS, AND THE TERMS OF TRADE

1140

- David Masclet, Charles Noussair, Steven Tucker, Marie-Claire Villeval, MONETARY AND NON-MONETARY PUNISHMENT IN THE VOLUNTARY CONTRIBUTIONS MECHANISM
- 1142 Charles Noussair, Stephane Robin, Bernard Ruffieux, DO CONSUMERS NOT CARE ABOUT BIOTECH FOODS OR DO THEY JUST NOT READ THE LABELS